Author's response to reviews

Title: Implementation of a self-management support approach (WISE) across a health system. A process evaluation explaining what did and didn't work for organisations, clinicians and patients

Authors:

Anne Kennedy (a.kennedy@soton.ac.uk)
Anne Rogers (a.e.rogers@soton.ac.uk)
Carolyn Chew-Graham (c.a.chew-graham@keele.ac.uk)
Thomas Blakeman (t.m.blakeman@manchester.ac.uk)
Robert Bowen (robert.a.bowen@manchester.ac.uk)
Caroline Gardner (caroline.j.gardner@manchester.ac.uk)
Victoria Lee (victoria.lee@manchester.ac.uk)
Rebecca Morris (Rebecca.morria@manchester.ac.uk)
Joanne Protheroe (j.protheroe@keele.ac.uk)

Version: 2 Date: 7 November 2013

Author's response to reviews:

We have now added a conclusion at the end of the text as instructed.
We have removed the first line on page 18 as the article referred to has not yet been published.
We have taken the opportunity to change the acronym PCT to PCMO as Primary Care Trusts (PCTs) no longer exist.